Author’s response to reviews

Title: Outcomes of thrombolytic therapy in acute ischemic stroke: mothership, drip-and-ship, and ship-and-drip paradigms

Authors:
Naruchorn Kijpaisalratana (naruchorn.k@chula.ac.th)
Aurauma Chutinet (aurauma@yahoo.com)
Wasan Akarathanawat (drwasan@outlook.com)
Pakkawan Vongvasinkul (pakkawan57@gmail.com)
Nijasri Suwanwela (nijasris@yahoo.com)

Version: 3 Date: 24 Jan 2020

Author’s response to reviews:
The revision has been made according to the editor comments.

1. The clear statement was added regarding the name of the IRB and the exemption of the consent form in the retrospective study.

2. Figure 2 was cited in the "clinical and safety outcomes."

3. All highlights were removed.